Author's response to reviews

Title: Human T cell Lymphotropic Virus Type 1- associated infective dermatitis in KwaZulu Natal, South Africa

Authors:

   Carol Hlela (carol.hlela@uct.ac.za)
   Natalie Graham (Natalie.Graham@scionresearch.com)
   Ahmed Bhigjee (bhigjee@ukzn.ac.za)
   Graham P Taylor (g.p.taylor@imperial.ac.za)
   Nonhlanhla P Khumalo (n.khumalo@uct.ac.za)
   Anisa Mosam (mosam@ukzn.ac.za)

Version: 2 Date: 18 February 2013

Author's response to reviews: see over
A statement to this effect had therefore been included and underlined in the revised manuscript to this effect that is available for review by the Editor of this Journal.

Response: Written and signed consent to publish the information was definitely obtained from the participants.

The use of images was obtained from the participants.

Consent for publication: The Editor sought clarity on whether written consent to publish this case series including the images in the manuscript to this effect that is available for review by the Editor of this Journal.

Thank you for the opportunity to revise the manuscripts and to publish this case series including the images.

17 January 2013

Dear Editor,

email: carol.hlela@uct.ac.za
MDCM, FCDerm, PhD
MD Dermatology
BMC Dermatology

Dr Carol Hlela
Facsimile: +27 21 689 1864
Telephone: +27 21 689 5502
South Africa
Cape Town
Red Cross Children's Hospital
Division of Dermatology